# Analyzing critical success factors for a successful transition towards circular economy through DANP approach

Shahbaz Khan, Annayath Maqbool and Abid Haleem Department of Mechanical Engineering, Jamia Millia Islamia, New Delhi, India, and Mohd Imran Khan

Department of Business, Lovely Professional University, Phagwara, India

# Abstract

**Purpose** – The successful transition towards the circular economy is a requirement of this technological era. The objective of this study is to identify critical success factors behind the successful transition to the circular economy.

**Design/methodology/approach** – A systematic literature review and experts' inputs are used to identify the major critical success factors of a successful transition towards a circular economy. Further, DANP is applied to discover the interrelationships among the critical success factors dimensions by analyzing causal relations between the relevant critical success factors. Additionally, the ranking of significant critical success factors for the successful transition towards the circular economy is determined.

**Findings** – This study has used the DANP methodology to understand the relationships among the five dimensions and the twenty-two critical success factors and identified the key influencing critical success factors through their global weights. Research findings suggest that "vision regard to a circular economy" has the highest weights, followed by "financial sustainability" which is regarded as the most important CSF.

**Research limitations/implications** – The outcomes of this work may help organizations on issues related to the transition of the linear economy towards circular economy, in understanding the degree of importance of the each critical success factors, based on which the organizations can formulate an effective strategy to systematically emphasize critical success factors as per their importance for the successful transition towards circular economy.

**Originality/value** – This study tries to explore and analyze critical success factors of the transition from the linear economy to a circular economy. Further, the finding of this study provides deeper insights into academia and mangers that helps to formulate their action plan for moving towards the circular economy.

**Keywords** Circular economy, Critical success factors, DANP, DEMATEL, Transition, Sustainability **Paper type** Research paper

# 1. Introduction

The world biodiversity is declining, and the world economy is expected to quadruple by 2050 (Wijkman and Skånberg, 2017). Resources like fossil fuels, metals, fertile land, and water are increasingly hard to get for the massively increasing population. The current trend in the economy is linear and is generally known as "take-make-use-dispose" model (Ghisellini *et al.*, 2016), which is continuing from the start of the industrial revolution (Ness, 2008). This linear model is not working well as the circumstances have changed. The Circular Economy (CE), primarily focusing on the reuse of new materials, can be the best alternative for the linear economic model (Singh and Ordoñez, 2016). "CE is a general term covering all activities that reduce, reuse and recycle materials in production, distribution, and consumption processes" (Blomsma and Brennan, 2017). The concept of "closing material loops" in the CE aims for the sensible utilization of natural resources and the reducing reusing and recycling of biological materials to extricate their maximum benefits with the least consumption (Jeng *et al.*, 2019). The idea of CE is eminent in the scholarly world, industries, and government, but its



Management of Environmental Quality: An International Journal Vol. 31 No. 3, 2020 pp. 505-529 © Emerald Publishing Limited 1477-7835 DOI 10.1108/MEQ-09-2019-0191



Received 16 September 2019 Revised 5 December 2019 31 December 2019 Accepted 31 December 2019

Transition to circular

economy via DANP

505

incorporation is still limited. Today only certain regions of a few countries such as China, Japan, Netherland, Germany, the United Kingdom are adopting CE. The shifting from a linear economy towards the CE is not smooth, and various hindrances come in the way for a successful transition towards CE.

Progress towards a CE involves significant changes in all the processes involved in the development of a product (Almut *et al.*, 2016; Nakajima, 2000). In the quest for the potential business, industrial organizations are effectively engaging the progress from linearity to circularity (Nordic Council of Ministers, 2015; Tukker, 2015). The concepts of reuse, remanufacturing, and recycling are being used for and reorganized, thus accentuating the adoption benefits, projecting the primary difficulties, and uncovering the barriers to be overcome as to achieve a CE.

The elements which facilitate the successful transition towards the CE need to be acknowledged. They represent crucial areas to be focused on the successful transition towards a CE but are insufficiently studied, thus constituting a gap in the research. Based on this gap, we have framed the research question of this study:

RQ1. What are the CSFs for the successful transition towards a CE?

Based on this question, the objective of this study is:

- (1) to identify the CSFs for a successful transition towards CE based on systematic literature review;
- (2) to evaluate and rank the identified CSFs using DANP method;
- (3) to understand the interrelationships of CSFs in the transition towards CE.

The rest of the paper is organized as follows: section 2 deals with literature review; section 3 provides the overview of the adopted research methodology; section 4 analyzes the data and reported results; section 5 discusses the results; section 6 provides the major research implications; finally, section 7 states the conclusion and future scope.

### 2. Literature review

#### 2.1 Circular economy and related studies

Korhonen *et al.* (2018) defined CE as "an economy constructed from societal productionconsumption systems that maximize the service produced from the linear nature-societynature material and energy throughput flow." It can further be accomplished by utilizing closed materials streams, inexhaustible energy sources, and cascading type of energy streams. CE constrains the throughput stream to a dimension that nature endures and uses ecological system cycles in monetary cycles by regarding their natural generation rates (Korhonen *et al.*, 2018). The main objective of CE is to maximize the utility and value of the products through the processes of repairing, reusing, refurbishing, and recycling with reducing waste generation. (Merli *et al.*, 2018). The concepts of "eco-design," "cleaner production," and "waste management" are supported by the CE by limiting the resource consumption (Zhu *et al.*, 2011). The limited use of virgin resources, reduced waste generation, growth in sustainability with environmental, financial and social improvement are the objectives of the CE (Yu *et al.*, 2015).

Today CE is seen as a solution or part of the solution to the environmental, social, and resource challenges (Panigrahi *et al.*, 2019). Therefore, many researchers, industry, and government have paid more attention and several studies related to the CE and the same were reported in the literature. These studies range from manufacturing to fashion industries in the context of the (re)design, manufacturing, supply chain management, and performance



MEQ

31.3

evaluation (Smith *et al.*, 2017). For instance, a framework is proposed to measure the performance of a green supply chain in the context of CE (Kazancoglu *et al.*, 2018). Mangla *et al.* (2018) identified and analyzed fifteen barriers related to the effective circular supply chain management in the context of an emerging economy. They find that the lack of environmental laws and tax policies and lack of preferential tax policies is the significant barriers. A similar study is also conducted by Farooque *et al.* (2019) in the context of China and they show that weak environmental regulations and lack of market preference/pressure are significant barriers to shift towards the CE.

In order to implement the CE, Kalmykova *et al.* (2018) provide the strategies that are relevant to different parts of the supply chain. Also, the implementation of CE is discussed through the business models for the CE and operations management in the CE environment (Lopes de Sousa Jabbour *et al.*, 2019). Vlajic *et al.* (2018) focus on the reverse logistic aspect of circular supply chains to achieve sustainable business practices. Masi *et al.* (2018) consider that the transition towards a CE is in progress and face some barriers such as investment cost and lack of awareness. They advocate the adoption of CE practices related to investment recovery, customer cooperation, and green purchasing. Govindan and Hasanagic (2018) identified through systematic review drivers, barriers, and practice towards the transition to CE. The finding of this study shows that CE can be encouraged by government policies, tax reduction, and strict governance.

#### 2.2 Critical success factors for the successful transition towards the circular economy

The CSF "is an element that is necessary for an organization to achieve its goal." According to Boynton and Zmud (1984), "CSFs are those few things that must go well to ensure success for a manager or an organization." CSFs are "the key factors/ enablers/ activities needed for the successful accomplishment of any business/ phenomenon to happen and necessary for an organization to achieve their goals" (Haleem *et al.*, 2012; Khan *et al.*, 2018). CSF represents those zones of a system that should be given continuous attention to achieved better performance (Sfakianaki, 2019). In this paper, CSFs of transition towards CE are identified and categorized into five different dimensions (see Table I).

#### 3. Research methodology

Analyzing the CSFs for the successful transition towards the CE is an intricate Multiple-Attribute Decision Making (MADM) process. For this research, the goal is the successful transition towards CE. Thus, to fulfill this objective, CSFs of the transition towards a CE needs identification and analysis. In order to do so, initially, twenty-seven CSFs are identified through the literature review of the CE, sustainable supply chain management, and green supply chain management. After identification of twenty-seven CSFs, the list is provided to a designated expert group to validate the relevance of these CSFs. Experts suggested dropping of four CSFs that do not seem appropriate in the context of a CE and also advised merging of two of them. In this manner, twenty-two CSFs for the successful transition towards CE are finalized. Further, experts also suggested to categorize CSFs into different categories based on the nature of the CSFs, and we have incorporated the same and categorized these twenty-two CSFs into five different categories. These factors are not independent; instead, they mutually influence each other. Hence, we require a method which could also recognize the interactions between the factors.

Majority of the methods which could address the MADM problems, such as Analytic Hierarchy Process (AHP), Technique for Order of Preference by Similarity to Ideal Solution



Transition to circular economy via DANP

507

MEQ 31.3	Dimensions	Critical success factor	Description	References
508	Organizational	A1. Management commitment and support	The management commitment and support such as proper training of the employees and suppliers have made an important influence on initiatives of organizations to implement the CE	Dubey <i>et al.</i> (2019); Stratan (2017)
		A2. Vision in regard to CE	Vision in terms of goals, objectives, targets about the CE should be clear to shift towards the CE	Pan <i>et al.</i> (2015)
		A3. Policies for CE practices	Public policies play a crucial role in creating an enabling condition (recycling policies, environmental laws, etc.) for a CE to develop and bloom	Maitre-Ekern and Dalhammar (2016); Genovese <i>et al.</i> (2017)
		A4. Business models	The circular business models will help the organizations in increasing differentiation, decreasing cost to serve and own, generating new revenues, and reducing risks and their impact on the rules of resource supply and demand	Mangla <i>et al.</i> (2018). Stratan (2017)
	Economical	B1. Financial sustainability	Financial sustainability is the proper management of risks and effectively using financial resources and maximizes the impact	Stratan (2017)
		B2. Capital investments	Capital investment refers to funds invested in a firm or enterprise for securing its business ideas and expanding their businesses for the future and lasting use	Wyman (2017)
		B3. Reuse of resources	CE aims at prosperity, and a possible increase in prosperity without this being at the expense of finite resources. The success of a CE is measured mainly by the re-use of resources.	Stratan (2017)
Table I.Critical success factorsfor the successfultransition towards CE		B4. Remanufacturing/ reuse cost	Remanufactured/recycled products should be cheaper than virgin materials	Lieder and Rashid (2016); Shahbazi <i>et al.</i> (2016)
with dimensions				(continued)



Transition to circular	References	Description	Critical success factor	Dimensions
economy via DANP	MacArthur (2015)	The circularity indicators methods and monitoring offer companies with a methodology and tools to measure the performance of a	C1. Methods, indicators, and monitoring	Technological
509	Bressanelli <i>et al.</i> (2018); Pagoropoulos <i>et al.</i> (2017)	product or company in a circular economic perspective A digital platform can enable rapid development of innovative products and services collaboratively. Digital tools can be the catalyst for optimizing manufacturing and	C2. Integration of CE with digital technology	
	Stratan (2017)	business processes across an enterprise to gear up for the CE For the successful transition towards a CE, the people involved in the different areas of the CE need to be expert in their respective fields	C3. The expertise of key people in their respective fields	
	Van Weelden <i>et al.</i> (2016)	Organizational skills and know- how are necessary for a business manager to grow the business in the CE. The technical and managerial skill of the process standardization and workflow is the prime importance to develop businesses in the CE	C4. Technical know-how and skill development	
	Stratan (2017).	"Sustainable innovation is the engine for future growth and is a catalyst for revolutionizing the art of business." The shifting from linear to circular business models will depend on the ability of an organization to innovate	C5. Ability to innovate	
	Genovese <i>et al.</i> (2017), Maqbool <i>et al.</i> (2020)	The complexity in the products makes it necessary for using technological resources for effective and efficient recovery and reuse of products	C6. Technological resources for CE implementation	
	Singh and Ordoñez (2016), Ghisellini <i>et al.</i> (2016)	Using the up-cycling process in the CE for enhancing the quality of the recycled products	C7. Quality preservation of reused material	
Table I.	(continued)			



MEQ	Dimensions	Critical success factor	Description	References
51,5	Environmental	D1. Eco-innovation	Eco-innovation is regarded as a key factor in the shifting from a linear to	De Jesus and Mendonna, 2017
510	_	D2. Eco-design	Eco-design is considered as a catalyst to switch from linear economy to a CE. Products are required to be designed both for circular loops and for revenue generation	More <i>et al.</i> (2016)
		D3. Cleaner production	CE involves practices which when follows to improve the ecological efficiency	Ghisellini <i>et al.</i> (2016), Lieder and Rashid (2016)
		D4. Legal and regulatory environment	Environmental protection is among the objectives of the CE for which it includes regulatory law. Need to designing of laws for a transition to sustainability	Stratan (2017)
	Social	E1. Public awareness of CE	Inspire and gather people to raise awareness of the CE, and opportunities using proven techniques for innovation and design	Stratan (2017); Lieder and Rashid (2016); MacArthur (2015)
		E2. Employment generation E3. Consumer perception towards used product	CE possess the potential to generate employment at the local level Consumers usually have negative perceptions about the remanufactured products and show reluctance in buying the	Schiller <i>et al.</i> (2017); Ilić and Nikolić (2016) Ghisellini <i>et al.</i> (2016); Van Weelden <i>et al.</i> (2016)
Table I.			refurbished products	

(TOPSIS), VlseKriterijumska Optimizacija I KompromisResenje (VIKOR) methods, assume that these factors are independent of each other (Haleem *et al.*, 2019; Asees Awan, and Ali, 2019). Baykasoğlu *et al.* (2013) claimed that this assumption is not realistic for many practical cases. Therefore, the Analytic Network Process (ANP) seemed suitable for this study in order to find the dependence and feedback among the criteria. However, the ANP method assumes that the network structure of the problem is known, and the factors have equal weights. Yang *et al.* (2008) stated that the network structure is often unknown, and the weight of each element is different. Moreover, the comparison matrices of ANP are too many and complicated.

ANP has become a significant approach for addressing the multiple-attribute decisionmaking problems successfully, for example, policy evaluation problems, risk analysis, and factor research problems. This study uses the integration of DEMATEL and ANP methods. DEMATEL is employed to obtain the influence levels of each factor over other factors (Khan *et al.*, 2019). This is in addition to the detection of complex relationships and developing of NRM for the criteria. The NRM is developed to show the significant relationship among the factors. The vertices of this NRM show the factor and arcs represent the directional relationship between these factors. The relationship between the factors can be unidirectional or bidirectional. The direction of the arrow from the *i*th factor to the *j*th factor shows the influence of the "factor *i*" on "factor *j*" (Khan *et al.*, 2018). This means that in the case of unidirectional relationship, the influential factor occurs at the tale side of the arrow while influenced factor lies on the head side. The bidirectional relationship is shown with the help of



a bidirectional arrow  $(\leftrightarrow)$ . If arc exists between the factors it represents, then there is significant relationship among factors. Figure 1 shows the proposed research framework.

### 3.1 DANP method

After the identification of the CSFs, we developed an appropriate questionnaire, which was sent to experts. Responses are collected about the effect of one factor over another. The questionnaire comprised of two pairwise comparison matrices, one matrix for the dimensions which are the organizational, economic, technological, environmental, and social and second matrix for the factors (CSFs) of successful transition towards CE. The experts were asked first to evaluate the relative importance between the two factors using the five-point linguistic scale (i.e., influence to high influence) to evaluate their influential strength. The expert's input is used to determine the degrees of influence of each criterion. This data will be used to build the NRM. The steps involved in the DANP method are as follows:



MEQ Step 1: Generation of the matrix for direct influence

31,3

512

The matrix acquired from the experts indicates the influence of factor i on factor j, represented by  $x^{k} = [x^{k}_{ij}]$ , where k is the kth expert.

Step 2: Generation of overall direct relation matrix

The average matrix  $A = [a_{ij}]$  and can be calculated using N experts from the expression (1)

$$a_{ij} = \frac{\sum_{k=1}^{N} x_{ij}^k}{N} \tag{1}$$

Step 3: Generation of normalized initial direct relation matrix B from expressions (2) and (3)

$$B = A.S \tag{2}$$

where 
$$S = \frac{1}{\max_{1 \le i \le n} \sum_{j=1}^{n} a_{ij}}$$
 (3)

Step 4: Generation of total relation matrix "T" from expression (4)

$$T = B.(I - B)^{-1}$$
(4)

where "*I*" is the identity matrix.

$$D_{1} \qquad D_{j} \qquad D_{m}$$

$$C_{11} \qquad \dots \qquad C_{1n_{1}} \qquad C_{j1} \qquad \dots \qquad C_{jn_{2}} \qquad C_{m1} \qquad \dots \qquad C_{mm_{n}}$$

$$D_{1} \qquad \stackrel{C_{11}}{\stackrel{C_{12}}{:}} \qquad \begin{bmatrix} T_{D}^{11} \qquad \dots \qquad T_{D}^{1j} \qquad \dots \qquad T_{D}^{1m} \\ \vdots \qquad \vdots \qquad \vdots \qquad \vdots \qquad \vdots \\ C_{1n_{1}} \\ C_{j1} \\ \vdots \qquad \vdots \qquad \vdots \qquad \vdots \qquad \vdots \qquad \vdots \\ C_{jn_{2}} \\ C_{m1} \\ \vdots \qquad \vdots \qquad \vdots \qquad \vdots \qquad \vdots \qquad \vdots \qquad \vdots \\ T_{D}^{j1} \qquad \dots \qquad T_{D}^{jj} \qquad \dots \qquad T_{D}^{jm} \\ \vdots \qquad \vdots \qquad \vdots \qquad \vdots \qquad \vdots \qquad \vdots \qquad \vdots \\ T_{D}^{m1} \qquad \dots \qquad T_{D}^{mj} \qquad \dots \qquad T_{D}^{mm} \end{bmatrix}$$

$$(5)$$

Step 5: Calculation of NRM based parameters from the expressions (6) and (7)

$$R_i = \sum_{j=1}^n t_{ij} \text{ for all } i \tag{6}$$



$$C_i = \sum_{i=1}^{n} t_{ij} \text{ for all } j$$
 (7) Transition to circular

The row sum of the *i*th row elements of matrix *T* is represented by  $R_i$  as shown in expression (6) which signifies the influence of factor "*i*" on the other factors. Similarly, the column sum of the *j*th column elements  $t_{ij}$  of matrix *T* is represented by  $C_i$  is shown in expression (7), and it signifies influence on factor *j* by other factors. The difference  $(R_i - C_i)$  demonstrates the net influence the factor *i* adds to the system (Sufiyan *et al.*, 2019).

Moreover, if " $R_i - G$ " is positive, factor *i* is a net cause, while factor *i* is a net receiver if " $R_i - C_i$ " is negative and " $R_i + C_i$ " the prominence of factor *i*. Based on the value of the "R + C" and "R - C", the network relation map is developed. Factors having the positive values of "R - C" are classified as the influential group, while the factors having negative values of the "R - C" is classified as an influenced group. The individual influence of one factor over another factor is represented by the arrow in the network relationship matrix. In order to develop the individual influence of one factor over another factor, a threshold value is chosen. This threshold value is computed by the average value of the total relation matrix and is obtained for dimensions and factor level. When the value of an element of the "total influence matrix" exceeds the threshold value, then they should be plotted with arrows in an NRM. This signifies the directional relationships.

Step 6: Generation of normalized total relation matrix

The total influence matrix includes  $T_{\rm D}$  based on dimensions as well as  $T_{\rm C}$  based on criteria. Normalizing  $T_{\rm D}$  and  $T_{\rm C}$  by Equations (8)–(11),  $T_{\rm D}$  <sup>norm</sup> and  $T_{\rm C}$  <sup>norm</sup> can be derived.

3.1.1 Normalization of TD.

$$T_{D}^{norm} = \begin{bmatrix} \frac{t_{D}^{11}}{r_{1}} & \cdots & \frac{t_{D}^{1j}}{r_{1}} & \cdots & \frac{t_{D}^{1m}}{r_{1}} \\ \vdots & \vdots & \vdots & \vdots & \vdots \\ \frac{t_{D}^{i1}}{r_{i}} & \cdots & \frac{t_{D}^{ij}}{r_{i}} & \cdots & \frac{t_{D}^{im}}{r_{i}} \\ \vdots & \vdots & \vdots & \ddots & \vdots \\ \frac{t_{D}^{m1}}{r_{m}} & \cdots & \frac{t_{D}^{mj}}{r_{m}} & \cdots & \frac{t_{D}^{mm}}{r_{m}} \end{bmatrix}$$
(8)

Where,  $r_1 = \sum_{j=1}^{m} t_D^{1m}$ ;  $r_i = \sum_{j=1}^{m} t_D^{ij}$ ;  $r_m = \sum_{j=1}^{m} t_D^{mm}$ 

The normalized total influence matrix will be

$$T_{D}^{norm} = \begin{bmatrix} t_{D}^{norm11} & \dots & t_{D}^{norm1j} & \dots & t_{D}^{norm1m} \\ \vdots & \vdots & \vdots & \vdots & \vdots \\ t_{D}^{norm11} & \dots & t_{D}^{norm1j} & \dots & t_{D}^{norm1m} \\ \vdots & \vdots & \vdots & \ddots & \vdots \\ t_{D}^{normm1} & \dots & t_{D}^{normmj} & \dots & t_{D}^{normmm} \end{bmatrix}$$
(9)

المنسارات

513

economy via DANP

WWW

)

3.1.2 Normalization of  $T_c$ . The total relation matrix ( $T_c$ ) is grouped into clusters according to some dimensions, e.g., in our study; there are five dimensions; accordingly, five clusters are made column-wise. Each element in a cluster is divided by the row sum of its cluster as shown in expression (10).

$$D_{1} \qquad D_{j} \qquad D_{m}$$

$$C_{11} \ \dots \ C_{1n_{1}} \qquad C_{j1} \ \dots \ C_{jn_{2}} \qquad C_{m1} \ \dots \ C_{mm_{n}}$$

$$D_{1} \qquad \begin{array}{c} C_{11} \\ C_{12} \\ \vdots \\ C_{1n_{1}} \\ \vdots \\ C_{1n_{1}} \\ \vdots \\ C_{j1} \\ C_{j1} \\ C_{j1} \\ C_{j1} \\ C_{jn_{2}} \\ C_{mn_{1}} \\ C_{mn_{2}} \\ \vdots \\ C_{mm_{n}} \\ \end{array} \qquad \begin{array}{c} \frac{t_{C}^{i1}}{r_{i1}} \\ \vdots \\ \vdots \\ \vdots \\ \vdots \\ \vdots \\ c_{mm_{n}} \\ \end{array} \qquad \begin{array}{c} \frac{t_{C}^{i1}}{r_{i1}} \\ \vdots \\ \frac{t_{C}^{i1}}{r_{i1}} \\ \cdots \\ \frac{t_{C}^{mj}}{r_{ij}} \\ \cdots \\ \frac{t_{C}^{mj}}{r_{mj}} \\ \cdots \\ \frac{t_{C}^{mm}}{r_{mm}} \\ \end{array} \qquad (10)$$

Step 7: Generation of unweighted supermatrix (W)

The matrix  $T_{\rm C}$  <sup>norm</sup> is obtained after the normalization of the total influence matrix. By transposing  $T_{\rm C}$  <sup>norm</sup>, we obtain the unweighted supermatrix as shown in expression (11).

Step 8: Generation of the weighted matrix  $(W^*)$ 

A weighted supermatrix, which is improved traditional ANP by using equal weights to make it appropriate for the real world, is generated by multiplying the unweighted matrix (W) with the transpose of  $T_{\rm D}$  norm shown in equation (12).



MEQ

31,3

514



Step 8: Obtaining the criteria weight

The global priority vector, which defines the influential weights  $w = (w_1, \dots, w_j, \dots, w_n)$  from  $\lim (w^*)^{\alpha}$  for the criteria is obtained by limiting the weighted supermatrix. The weighted Supermatrix is raised to large power ( $\alpha$ ) until it converges and becomes a long-term stable super-matrix.

# 4. Results

This study assesses the CSFs for the successful transition towards a CE. The CSFs were identified based on extensive literature review and then finalized through the discussion with experts. In this paper, twenty-two CSFs were identified and were clustered into five dimensions shown in Table I.

# 4.1 Data collection for NRM

After finalizing the CSFs for the successful transition towards a CE, the questionnaire was developed and sent to experts to access the influential strength of CSFs. The questionnaire contained two pairwise comparison matrices which are expected to be filled by experts. The first comparison matrix contains the dimensions which are organizational, economic, technological, environmental, and social (refer to Table AI). The second comparison matrix contained the twenty-two CSFs. The experts were asked to give their responses about the relative importance of the two factors using the linguistic scale to evaluate their influential strength. The expert's responses are used to develop the NRM using equation (1)–(5).

#### 4.2 Network relations map

The total influence matrix can be calculated using the expert's responses by applying equations (1–4). After that, the influence of each dimension as well as CSFs are determined and these are presented in Table II and Table III, respectively.

Table II shows the central degrees (i.e., R+C) of the five dimensions. Among the five diminutions, the organizational dimension (9.273) has the highest central degree, followed by economic dimension (8.659). The order of importance of dimensions based on central degree is organizational (9.272) > economical (8.659) > technological (8.448) > environmental (7.096) > social (6.262). Organizational, economic, and environmental dimensions have positive values of (R-C), which implies that they influence the other dimensions and can be categorized under cause group. The other dimensions, technical and social, have negative of

CSFs	R	С	R+C	R-C	
A	4.858	4.413	9.272	0.445	Table II
B	4.492	4.167	8.659	0.325	
C	3.648	4.8	8.448	-1.152	The prominence $(R+C)$
D	3.852	3.244	7.096	0.608	and net effect $(R-C)$ on
Е	3.018	3.244	6.262	-0.227	dimensions

MEQ 31.3	CSFs	Ri	Ci	$R_{\rm i}+C_{\rm i}$	$R_{\rm i}-C_{\rm i}$
51,0	Δ1	0.417	0.452	0.869	_0.035
	A2	0.531	0.402	1 049	0.012
	A3	0.454	0.401	0.855	0.012
	A4	0.472	0.503	0.975	-0.031
	R1	0.328	0.306	0.634	0.001
516	B2	0.304	0.278	0.582	0.022
	B3	0.226	0.323	0.548	-0.027
	B4	0.258	0.209	0.467	0.048
	C1	0.284	0.259	0.543	0.024
	C2	0.27	0.277	0.547	-0.008
	C3	0.277	0.235	0.512	0.042
	C4	0.224	0.244	0.468	-0.02
	C5	0.267	0.243	0.511	0.024
	C6	0.239	0.34	0.579	-0.101
	C7	0.299	0.26	0.559	0.038
	D1	0.165	0.177	0.341	-0.012
	D2	0.157	0.154	0.311	0.003
Table III	D3	0.193	0.164	0.357	0.029
Prominence $(R+C)$	D4	0.123	0.144	0.267	-0.021
with net effect $(R-C)$	E1	0.038	0.056	0.094	-0.018
on Critical Success	E2	0.056	0.031	0.088	0.025
Factors	E3	0.04	0.047	0.087	-0.007

(R-C), which implies that they are being influenced by other dimensions and can be placed under the effect group.

Table III shows the prominence of the CSFs in which "vision in regards of CE (1.049)" has the highest central degree followed by "business models (0.975)" in the organizational dimension. Among the organizational CSFs, "vision in regard to CE" and "policies for CE practices" have positive values of (R-C) means they directly influence the other CSFs in the organizational dimension.

In the economic dimension, the most prominent CSF is "financial sustainability" having the highest values of (R+C), and the most influential is "remanufacturing/ reuse cost" with the highest value of (R-C) and most influenced CSF is "reuse of resources." Among the technological CSFs, "technological resources for CE implementation" is the prominent CSFs having the highest degree of prominence and "expertise of key people in their respective fields" is the most influential CSF. In the environmental dimension, "cleaner production" has the highest central value, and "legal and regulatory environment" is the most influenced CSF. In the social dimension, the highest value of "R+C" belongs to "public awareness of CE" and "employment generation" is among the cause factors having positive values of (R-C) and "public awareness" of CE and "consumer perception towards used product" are influenced CSFs having negative values of (R-C). Based on Table II and Table III, the network relations map is developed and shown below in Figure 2 (a)–(f).

### 4.3 Determining global weight of each critical success factor

Total influence matrix of demotions generated using equation (1)–(6), and it is shown in Table AI. Similarly, the total influence matrix of the CSFs is developed, and it is shown in Table AII. Further, the total influence matrix of the demotions and CSFs are transformed into normalized total influence matrix using equation (6)–(9), and the same is shown in Table AIII and AIV. The normalized total influence matrix is transposed to generate the unweighted supermatrix (*W*) using equation (11), and it is shown in Table AV. The weighted supermatrix is generated by multiplying the unweighted matrix (*W*) with the transpose of  $T_D$  norm (refer to





Table AIII) using equation (12) as shown in Table AVI. Finally, the weighted supermatrix is raised to the power of 15 to generate a stable limiting matrix (refer to Table AVII), from which we get the global weights of each CSF.

CSFs are ranked according to global weights, here vision in regard to CE has the highest weight and is followed by financial sustainability. Table IV shows the ranking order of various CSFs.

## 5. Discussion on results

This study focuses on the CSFs of the transition of the linear economy towards the CE. The result shows that the organizational dimension has the highest value of centrality, which implies that the organizational dimension needs to be given higher consideration for the successful transition towards the CE. Environmental, organizational, and economic dimensions have the positive values of "R-C" and are placed in an influential group while social and technological dimensions are having negative values and placed in the influenced group. It is evident in Figure 2 (a) that environmental, organizational, economic dimensions



MEO				
		Local weights based	Global weights	
31,3	CSFs	on global weights	based on DANP	Rank
	A Organizational	0.267		
	A1 Management commitment and support	0.246	0.066	7
	A2 Vision in regard to CE	0.277	0.074	1
	A3 Policies for CE practices	0.211	0.056	12
518	A4 Business models	0.267	0.071	3
010	B Economical	0.251	0.071	0
	B1 Financial sustainability	0.284	0.072	2
	B2 Capital investments	0.259	0.065	8
	B3 Reuse of resources	0.203	0.068	4
	B4 Remanufacturing/ reuse cost	0.184	0.046	15
	C Technological	0.300	01010	10
	C1. Methods, indicators, and monitoring	0.134	0.040	20
	C2. Integration of CE with technology	0.142	0.043	17
	C3. The expertise of key people in their respective fields	0.137	0.041	19
	C4. Technical know-how and skill development	0.117	0.035	22
	C5. Ability to innovate	0.127	0.038	21
	C6. Technological resources for CE implementation	0.191	0.057	11
	C7. Quality preservation of reused material	0.153	0.046	16
	D. Environmental	0.197		
	D1. Eco-innovation	0.249	0.049	13
	D2. Eco-design	0.208	0.041	18
	D3. Cleaner production	0.303	0.060	10
	D4. Legal and regulatory environment	0.239	0.047	14
Table IV.	E. Social	0.195		
Global and local	E1. Public awareness of CE	0.347	0.068	5
weights of dimensions	E2. Employment generation	0.342	0.067	6
and CSFs	E3. Consumer perception towards used product	0.311	0.061	9

are influential dimensions, and they influenced the social and technological dimensions. The environmental dimension is the most influential dimension, as it influences all the other four dimensions, while the technological dimension is the highest influenced dimensions. It implies that the management and policy planners should focus on the influencing dimensions (i.e., environmental, organizational, economic) for the successful transition towards CE.

As shown in Figure 2 (b), the most influential CSF is the "policies for CE practices" among the organizational factors. As "policies for CE practices" influence, the other three CSFs and "business model" is influenced by other remaining three CSFs. It is interesting to note that the three CSFs related to economic dimension belong to the influential group. The "financial sustainability," "capital investments," and "remanufacturing/reuse cost" influenced the "reuse of resources" as shown in Figure 2 (c). The four CSFs related to the technological dimension belong to the influential group, and the remaining three belongs to the influenced group. It is clear from Figure 2 (d) that "expertise of key people in their respective fields" is the highest influential CSFs among all the CSFs related to technological dimension. As Figure 2 (e) shows that "cleaner production" is the most influential CSF which influenced the remaining three CSFs. Figure 2 (f) represents that "employment generation" belongs to the influential group, and it influences the "public awareness of CE" and "consumer perception towards a used product."

Overall, the network relations map shows that the causal relationship among the CSFs, which specifies that "policies for CE practices," "remanufacturing/reuse cost," "expertise of key people in their respective fields," "cleaner production," and "employment generation" as the main factors to drive other CSFs in the respective dimensions. This signifies that the **policymakers**, **including governments**, must improve their policies for the successful



transition and expertise knowledge should be used for the effective development in any field of the CE. Cleaner production and employment generation should be one of the main aims for the transition towards the CE.

Table IV shows the ranking of CSFs for the successful transition towards the CE. The "vision regarding CE" at the top of the list among the all identified CSFs, and this implies that there should be a clear vision in terms of goals and objectives, i.e., what and why of CE and its implementation. The "vision regarding CE" is followed by "financial sustainability" and ranked second. Financial resources play an essential role in any development which must be appropriately used for the successful transition. The emphasis should be laid on the "business models" (ranked third) and its innovation as they are regarded as the main actors in the transition towards a CE. The circular business models will help the organizations in increasing differentiation, decreasing cost to serve and own, generating new revenues and reducing risks and also their impact on the rules of resource supply and demand. The "business models" are followed by the "reuse of resources" (ranked fourth), which stresses on the effective utilization of used products. The success of the CE, or the way towards it, is, therefore, to be measured mainly by the degree of decoupling from economic activity and the re-use of resources. The "public awareness of CE" is ranked five, which implies that raising awareness among the masses will ease the transition towards the CE. The public should be encouraged and inspired by the positives in implementing the CE.

Other CSFs like "employment generation," "management commitment and support," "capital investments," "consumer perception towards used product," and "cleaner production" are ranked sixth, seventh, eighth, ninth, and tenth respectively based on global weights. A CE can be a source of employment generation by effectively incorporating the CE practices in existing organizations. The repair, reuse, and recycle approach of the CE offers the opportunity to bring back professions and jobs that had disappeared in recent times, especially in emerging economies. The leadership skills and commitment of top management has a significant impact on the initiatives of an organization towards the transaction of the linear economy into CE. Another important CSFs is "capital investment" which plays an important role in the transition as the companies making the transition to CE principles have unique financial requirements. The negative consumer perception towards the remanufactured products and their reluctance in buying the refurbished products makes it an important CSF to be focused on. Cleaner production results in ecological efficiency, influence the people to prefer the CE. It implies that these CSFs' also play a role in the successful transition towards the CE.

### 6. Research implications

This study provides managerial as well as theoretical implications which are presented in this section.

#### 6.1 Managerial implications

The outcomes of this work may help organizations on issues related to the transition of the linear economy towards CE, in understanding the degree of importance of each CSF. The causal interrelationship among the CSFs assists the management to understand the influence of one CSF over others which can help in the formulation of an effective strategy for the successful transition towards CE. Further, the ranking of the CSFs might be helpful to priorities their action plan by focusing only on the highest-ranked CSF in order to move towards CE. This study might be helpful for the managers who are trying the transition towards CE in a more practical way. The findings of this research could benefit the managers interested in developing and maintaining organizations based on CE principles. Based on this study, one could work for improvements in the CSFs according to their weights, and measure the successful transition towards a CE.



6.2 Theoretical implications

MEQ

31.3

520

The twenty-two CSFs clustered into five dimensions for the successful transition towards a CE given in this study provides theoretical insinuation for the scholarly discussion on the CE. The findings of this research will also help academia in order to develop the understanding and the importance of the transition towards the CE. Research related to the CE is in its initial stage, and this study will motivate academician to work in this area to obtain sustainability. The recognized CSFs of the transition towards CE will help academia to develop the conceptual framework for the implementation of CE practices. This study is also beneficial for academia, researchers, and industry practitioners to understand more about the transition towards CE in developing economies such as India, Chain, and other countries of south–east Asia. The framework adopted in this study can also be used for the assessment of the implementation of CE practices.

# 7. Conclusion, limitation, and future scope

Although the transition towards a CE has many benefits, for example, it limits the virgin resource extraction, focuses more on sustainable development and may increase the job opportunities. Therefore, the elements which facilitate the successful transition from a linear economy to CE need to be acknowledged. This study undertakes the issue of transition from a linear economy to CE. In order to do so, the CSFs for the successful transition towards the CE are identified through the literature review and finalized through the discussion with experts. Further, these CSFs are classified into five different demotions based on their similarities. After finalizing the dimensions and their associated CSFs, it was analyzed using the DANP approach. These CSFs were ranked, and the result is discussed with the experts for gaining useful insights. Some of the significant insights are provided in the discussion section. This study has used DEMATEL to describe the relationships between the five dimensions and the twenty-two CSFs and constructed an NRM. Secondly, the DANP method is used for quantitative analysis in which global weights are calculated, and the key influencing CSFs identified. The results illustrate that "vision in regard to CE" has the highest weights, followed by "financial sustainability" and are regarded as the most important CSFs for the successful transition towards CE. This study provides opportunities for advanced research in the emerging field of the CE.

The limitation of this study is that it is based on the review of the literature and experts' input. Thus, there is a chance to overlook some CFSs while conducting the literature review. Second, the prioritization is based on the expert input that might be biased towards their working level. The expert input is taken in the subjective form, which may bring some error in weights of the CSFs. This study is only limited to the analyzing the CSFs of the transition towards CE and does not consider the assessment of the adoption of CE practices or measurement of a successful transition toward CE. These limitations open the door for future research. In future, a more comprehensive literature review can be done in order to retain more CSFs. Further, fuzzy or grey theory can be integrated with the DANP method to deal with the subjectivity in the expert's input. The performance measurement of an organization towards a successful transition to CE can be assessed in future studies. From the methodological perspective, these CSFs can be evaluated by other integrated MCDM technique such as fuzzy DANP, fuzzy best-worst method, grey TOPSIS, and so on.

### References

Almut, R., De Schoenmakere, M. and Jeroen, G. (2016), "Circular economy in Europe: developing the knowledge base", in Ullstein, B, Saunders, P. and de Mattos, H. (Eds.), *European Environment Agency*, Office of the European Union, Luxembourg.

Asees Awan, M. and Ali, Y. (2019), "Sustainable modeling in reverse logistics strategies using fuzzy MCDM", *Management of Environmental Quality: An International Journal*, Vol. 30 No. 5,



Baykasoğlu, A., Kaplanoğlu, V., Durmuşoğlu, Z. and Şahin, C. (2013), "Integrating fuzzy DEMATEL and fuzzy hierarchical TOPSIS methods for truck selection", *Expert Systems with Applications*, Vol. 40 No. 3, pp. 899-907.

- Blomsma, F. and Brennan, G. (2017), "The emergence of circular economy: a new framing around prolonging resource productivity", *Journal of Industrial Ecology*, Vol. 21 No. 3, pp. 603-614.
- Boynton, A. and Zmud, R. (1984), "An assessment of critical success factors", Sloan Management Review, Vol. 25, pp. 17-27.
- Bressanelli, G., Perona, M. and Saccani, N. (2018), "Challenges in supply chain redesign for the Circular Economy: a literature review and a multiple case study", *International Journal of Production Research*, Vol. 57 No. 23, pp. 7395-7422.
- De Jesus, A. and Mendonna, S. (2017), "Lost in transition? drivers and barriers in the eco-innovation road to the circular economy", SSRN Electronic Journal, Vol. 145, pp. 75-89, doi: 10.2139/ssrn. 3038887.
- Dubey, R., Gunasekaran, A., Childe, S., Papadopoulos, T. and Helo, P. (2019), "Supplier relationship management for circular economy", *Management Decision*, Vol. 57 No. 4, pp. 767-790.
- Farooque, M., Zhang, A. and Liu, Y. (2019), "Barriers to circular food supply chains in China", Supply Chain Management: International Journal, Vol. 24 No. 5, pp. 677-696, doi: 10.1108/scm-10-2018-0345.
- Genovese, A., Acquaye, A., Figueroa, A. and Koh, S. (2017), "Sustainable supply chain management and the transition towards a circular economy: Evidence and some applications", *Omega*, Vol. 66, pp. 344-357.
- Ghisellini, P., Cialani, C. and Ulgiati, S. (2016), "A review on circular economy: the expected transition to a balanced interplay of environmental and economic systems", *Journal of Cleaner Production*, Vol. 114, pp. 11-32.
- Govindan, K. and Hasanagic, M. (2018), "A systematic review on drivers, barriers, and practices towards circular economy: a supply chain perspective", *International Journal of Production Research*, Vol. 56 Nos 1-2, pp. 278-311.
- Haleem, A., Khan, S. and Khan, M. (2019), "Traceability implementation in food supply chain: a grey-DEMATEL approach", *Information Processing in Agriculture*, Vol. 6 No. 3, pp. 335-348.
- Haleem, A., Sushil, Qadri, M. and Kumar, S. (2012), "Analysis of critical success factors of world-class manufacturing practices: an application of interpretative structural modelling and interpretative ranking process", *Production Planning & Control*, Vol. 23 Nos 10-11, pp. 722-734.
- Ilić, M. and Nikolić, M. (2016), "Drivers for development of circular economy a case study of Serbia", *Habitat International*, Vol. 56, pp. 191-200.
- Jeng, S., Lin, C., Tseng, M. and Jantarakolica, T. (2019), "Cradle-to-cradle zero discharge production planning system for the pulp and paper industry using a fuzzy hybrid optimization model", *Management of Environmental Quality: An International Journal*, doi: 10.1108/meq-06-2019-0120.
- Kalmykova, Y., Sadagopan, M. and Rosado, L. (2018), "Circular economy from review of theories and practices to development of implementation tools", *Resources, Conservation and Recycling*, Vol. 135, pp. 190-201, doi: 10.1016/j.resconrec.2017.10.034.
- Kazancoglu, Y., Kazancoglu, I. and Sagnak, M. (2018), "A new holistic conceptual framework for green supply chain management performance assessment based on circular economy", *Journal of Cleaner Production*, Vol. 95, pp. 1282-1299.
- Khan, S., Haleem, A., Khan, M., Abidi, M. and Al-Ahmari, A. (2018), "Implementing traceability systems in specific supply chain management (SCM) through critical success factors (CSFs)", *Sustainability*, Vol. 10 No. 2, p. 204.
- Khan, S., Imran Khan, M. and Haleem, A. (2018), "Towards Effective Management of Cold Chain: A DEMATEL Approach", *IOP Conference Series: Materials Science and Engineering*, Vol. 404, p. 012019.



521

Transition

to circular

via DANP

economy

MEQ	Khan, S., Khan, M. and Haleem, A. (2019), "Evaluation of barriers in the adoption of halal certification: a fuzzy DEMATEL approach", <i>Journal of Modelling in Management</i> , Vol. 14 No. 1, pp. 153-174.
51,5	Korhonen, J., Honkasalo, A. and Seppälä, J. (2018), "Circular economy: the concept and its limitations", <i>Ecological Economics</i> , Vol. 143, pp. 37-46.
	Lieder, M. and Rashid, A. (2016), "Towards circular economy implementation: a comprehensive review in context of manufacturing industry", <i>Journal of Cleaner Production</i> , Vol. 115, pp. 36-51.
522	<ul> <li>Lopes de Sousa Jabbour, A., Rojas Luiz, J., Rojas Luiz, O., Jabbour, C., Ndubisi, N., Caldeira de Oliveira,</li> <li>J. and Junior, F. (2019), "Circular economy business models and operations management",</li> <li><i>Journal of Cleaner Production</i>, Vol. 235, pp.1525-1539, doi: 10.1016/j.jclepro.2019.06.349.</li> </ul>
	MacArthur, E. (2015), "Towards a circular economy: business rationale for an AcceleratedTransition", available at: https://www.ellenmacarthurfoundation.org/assets/downloads/TCE_Ellen-MacArthur- Foundation_9-Dec-2015.pdf.
	Maitre-Ekern, E. and Dalhammar, C. (2016), <i>Regulating Planned Obsolescence: A Review of Legal Approaches to Increase Product Durability and Reparability in Europe</i> , Review of European, Comparative & International Environmental Law, Wiley online library, Vol. 25 No. 3, pp. 378-394.
	Mangla, S., Luthra, S., Mishra, N., Singh, A., Rana, N., Dora, M. and Dwivedi, Y. (2018), "Barriers to effective circular supply chain management in a developing country context", <i>Production</i> <i>Planning &amp; Control</i> , Vol. 29 No. 6, pp. 551-569.
	Maqbool, A., Khan, S., Haleem, A. and Khan, M. I. (2020), "Investigation of Drivers Towards Adoption of Circular Economy: A DEMATEL Approach", <i>Lecture Notes in Mechanical Engineering</i> <i>Recent Advances in Mechanical Engineering</i> , pp. 147–160, doi: 10.1007/978-981-15-1071-7_14.
	Masi, D., Kumar, V., Garza-Reyes, J. and Godsell, J. (2018), "Towards a more circular economy: exploring the awareness, practices, and barriers from a focal firm perspective", <i>Production Planning &amp; Control</i> , Vol. 29 No. 6, pp. 539-550.
	Merli, R., Preziosi, M. and Acampora, A. (2018), "How do scholars approach the circular economy? a systematic literature review", <i>Journal of Cleaner Production</i> , Vol. 178, pp. 703-722.
	Moreno, M., De los Rios, C., Rowe, Z. and Charnley, F. (2016), "A conceptual framework for circular design", <i>Sustainability</i> , Vol. 8 No. 9, p. 937.
	Nakajima, N. (2000), "A vision of industrial ecology: state-of-the-art practices for a circular and service- based economy", <i>Bulletin of Science, Technology &amp; Society</i> , Vol. 20 No. 1, pp. 54-69.
	Ness, D. (2008), "Sustainable urban infrastructure in China: towards a Factor 10 improvement in resource productivity through integrated infrastructure system", <i>International Journal of Sustainable Development &amp; World Ecology</i> , Vol. 15, pp. 288-301, doi: 10.3843/SusDev.15.4:2.
	Nordic Council of Ministers (2015), Moving towards a Circular Economy -successful Nordic Business Models.
	Pagoropoulos, A., Pigosso, D. and McAloone, T. (2017), "The emergent role of digital technologies in the circular economy: a review", <i>Procedia CIRP</i> , Vol. 64, pp. 19-24.
	Pan, S., Du, M., Huang, I., Liu, I., Chang, E. and Chiang, P. (2015), "Strategies on implementation of waste-to-energy (WTE) supply chain for circular economy system: a review", <i>Journal of Cleaner</i> <i>Production</i> , Vol. 108, pp. 409-421.
	Panigrahi, S., Bahinipati, B. and Jain, V. (2019), "Sustainable supply chain management", Management of Environmental Quality: An International Journal, Vol. 30 No. 5, pp. 1001-1049.
	Schiller, G., Müller, F. and Ortlepp, R. (2017), "Mapping the anthropogenic stock in Germany: metabolic evidence for a circular economy", <i>Resources, Conservation and Recycling</i> , Vol. 123, pp. 93-107.
	Sfakianaki, E. (2019), "Critical success factors for sustainable construction: a literature review", Management of Environmental Quality: An International Journal, Vol. 30 No. 1, pp. 176-196.
	Shahbazi, S., Wiktorsson, M., Kurdve, M., Jönsson, C. and Bjelkemyr, M. (2016), "Material efficiency in manufacturing: Swedish evidence on potential, barriers and strategies", <i>Journal of Cleaner</i> <i>Production</i> , Vol. 127, pp. 438-450.



- Singh, J. and Ordoñez, I. (2016), "Resource recovery from post-consumer waste: important lessons for the upcoming circular economy", *Journal of Cleaner Production*, Vol. 134, pp. 342-353.
- Smith, P., Baille, J. and McHattie, L. (2017), "Sustainable Design Futures: an open design vision for the circular economy in fashion and textiles", *The Design Journal*, Vol. 20, pp. S1938-S1947, doi: 10.1080/14606925.2017.1352712.
- Stratan, D. (2017), "Success factors of sustainable social enterprises through circular economy perspective", Visegrad Journal on Bioeconomy and Sustainable Development, Vol. 6 No. 1, pp. 17-23.
- Sufiyan, M., Haleem, A., Khan, S. and Khan, M. (2019), "Evaluating food supply chain performance using hybrid fuzzy MCDM technique", *Sustainable Production and Consumption*, Vol. 20, pp. 40-57.
- Tukker, A. (2015), "Product services for a resource-efficient and circular economy a review", *Journal* of Cleaner Production, Vol. 97, pp. 76-91.
- Van Weelden, E., Mugge, R. and Bakker, C. (2016), "Paving the way towards circular consumption: exploring consumer acceptance of refurbished mobile phones in the Dutch market", *Journal of Cleaner Production*, Vol. 113, pp. 743-754.
- Vlajic, J., Mijailovic, R. and Bogdanova, M. (2018), "Creating loops with value recovery: empirical study of fresh food supply chains", *Production Planning & Control*, Vol. 29 No. 6, pp. 522-538.
- Wijkman, A. and Skånberg, K. (2017), "The Circular Economy and Benefits for Society: jobs and climate clear winners in an economy based on renewable energy and resource efficiency, available at: https://www.clubofrome.org/wp-content/uploads/2016/03/The-Circular-Economyand-Benefitsfor-Society.pdf, (accessed 23 March 2019).
- Wyman, O. (2017), "Supporting the circular economic transition", available at: https://www. oliverwyman.com/content/dam/oliverwyman/v2/publications/2017/sep/CircularEconomy\_print. pdf (accessed 05 March 2019).
- Yang, Y.O., Shieh, H.M., Leu, J.D. and Tzeng, G.H. (2008), "A novel hybrid MCDM model combined with DEMATEL and ANP with applications", *International Journal of Operation Research*, Vol. 5 No. 3, pp. 160-168.
- Yu, F., Han, F. and Cui, Z. (2015), "Evolution of industrial symbiosis in an eco-industrial park in China", Journal of Cleaner Production, Vol. 87, pp. 339-347.
- Zhu, Q., Geng, Y. and Lai, K. (2011), "Environmental supply chain cooperation and its effect on the circular economy practice-performance relationship among Chinese manufacturers", *Journal of Industrial Ecology*, Vol. 15 No. 3, pp. 405-419.

### Further reading

Geissdoerfer, M., Savaget, P., Bocken, N. and Hultink, E. (2017), "The Circular Economy – a new sustainability paradigm?", *Journal of Cleaner Production*, Vol. 143, pp. 757-768.

#### **Corresponding author**

Abid Haleem can be contacted at: ahaleem@jmi.ac.in

#### Appendix

	E	D	С	В	А	
	0.801	0.801	1.244 1.166	1.110	0.902	A
Table Total relation mate	0.614	0.614	0.743	0.811	0.865	Č D
Dimens	0.410	0.553	0.745	0.610	0.70	E



MEQ 31,3         524           5         1
53.30         23.4         100         23.0         100         23.0         100         23.0         100         23.0         100         23.0         100         23.0         100         23.0         100         23.0         100         23.0         100         23.0         100         23.0         100         23.0         100         23.0         100         23.0         100         23.0         100         23.0
5224         III         III         III         III         IIII         IIIII         IIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIII
5224         H
1         3.1
1         A3         A1         B1         B         B1         C1         C3         C4
N         M         B1         B2         B         H         C         D
1         A3         A         B1         B2         B3         B4         C1         C3         C4
1         1         1         2         1         2         1         2
1         A3         A4         B1         B2         B3         A4         C </td
1         A3         A4         B1         B2         B3         B4         C1         C2         C3           8         0.065         0.126         0.097         0.097         0.112         0.075         0.073         0.073           11         0.113         0.113         0.113         0.113         0.113         0.110         0.104         0.073 <td< td=""></td<>
No         Al         B1         B2         B3         A4         B1         C         C         C         C           8         0.095         0.136         0.136         0.122         0.129         0.076         0.096         0.130         0.104           8         0.095         0.136         0.136         0.132         0.1128         0.007         0.007         0.006         0.096         0.100         0.006         0.004         0.006
x         x3         x4         B1         B2         B3         B4         c1         c2           x         x3         x4         B1         B2         B3         B4         c1         c3           x         0069         0.130         0.132         0.112         0.076         0.097         0.013           x         0069         0.130         0.122         0.129         0.013         0.010         0.075           x         0069         0.130         0.122         0.126         0.091         0.101         0.103           x         0.0069         0.130         0.122         0.126         0.091         0.013         0
R         A3         A4         B1         B2         B3         B4         C1           2         A3         A4         B1         B2         B3         B4         C1           2         0.095         0.126         0.097         0.097         0.129         0.097           2         0.139         0.130         0.139         0.139         0.139         0.097           2         0.096         0.130         0.139         0.139         0.139         0.017         0.096           2         0.139         0.151         0.139         0.139         0.139         0.017         0.091           2         0.096         0.130         0.131         0.131         0.131         0.017         0.091           2         0.096         0.130         0.111         0.129         0.076         0.091         0.077         0.091           2         0.081         0.0149         0.055         0.061         0.096         0.092         0.091           3         0.018         0.0149         0.055         0.061         0.096         0.092         0.092         0.092           3         0.018         0.0149         0.052 </td
R         A3         A4         B1         B2         B3         B4           8         0.095         0.126         0.097         0.0129         0.076           8         0.134         0.131         0.1129         0.076         0.076           8         0.095         0.130         0.1229         0.0129         0.076           8         0.133         0.113         0.112         0.1129         0.076           9         0.0131         0.113         0.113         0.0126         0.081         0.046           1         0.103         0.0255         0.0111         0.112         0.0126         0.081         0.046           1         0.013         0.0126         0.0131         0.112         0.0126         0.0167         0.026         0.0161         0.046         0.077           1         0.016         0.0176         0.0067         0.0267         0.0267         0.028         0.0141         0.040           1         0.008         0.0110         0.077         0.0267         0.041         0.040           1         0.008         0.014         0.096         0.067         0.028         0.041         0.043
:     A3     A4     B1     B2     B3       2     0.095     0.126     0.097     0.0122     0.1292       2     0.0134     0.151     0.030     0.097     0.0132       2     0.0095     0.133     0.1118     0.1123     0.1132       2     0.0095     0.133     0.0122     0.0131       3     A4     B1     B2     B3       4     0.013     0.0123     0.122     0.0132       1     0.103     0.0095     0.014     0.0095       9     0.081     0.1014     0.0073     0.0064       0     0.081     0.1012     0.075     0.064       0     0.081     0.1022     0.064     0.014       0     0.081     0.1022     0.066     0.065       0     0.081     0.1022     0.064     0.067       0     0.083     0.1022     0.064     0.067       0     0.083     0.1022     0.064     0.067       0     0.083     0.1022     0.064     0.067       0     0.083     0.066     0.077     0.063       0     0.083     0.064     0.064     0.066       0     0.083     0.066
:     A3     A4     B1     B2       :     A3     A4     B1     B2       :     0.095     0.1512     0.097     0.097       :     0.046     0.130     0.113     0.126       :     0.046     0.130     0.130     0.126       :     0.066     0.130     0.130     0.097       :     0.066     0.130     0.111     0.126       :     0.061     0.120     0.064     0.091       :     0.063     0.121     0.0126     0.061       :     0.061     0.120     0.064     0.064       :     0.066     0.120     0.065     0.061       :     0.061     0.102     0.065     0.061       :     0.066     0.099     0.077     0.065       :     0.066     0.090     0.064     0.066       :     0.064     0.069     0.077     0.065       :     0.062     0.061     0.066     0.061       :     0.068     0.077     0.065     0.065       :     0.068     0.077     0.065     0.065       :     0.068     0.077     0.065     0.065       :     0.069     0.0
A3         A4         B1           28         0.095         0.126         0.097           7         0.069         0.130         0.130           7         0.069         0.130         0.130           7         0.069         0.130         0.095           7         0.069         0.130         0.0130           7         0.069         0.130         0.005           7         0.069         0.130         0.005           8         0.0310         0.007         0.0134           9         0.081         0.104         0.086           11         0.088         0.110         0.075           9         0.081         0.104         0.086           11         0.078         0.0104         0.086           11         0.079         0.076         0.076           11         0.078         0.0104         0.067           11         0.079         0.0102         0.077           11         0.079         0.0044         0.067           11         0.079         0.0102         0.075           11         0.079         0.0044         0.067
A3         A4           8         0.095         0.126           8         0.065         0.126           9         0.081         0.103           9         0.081         0.103           9         0.081         0.102           9         0.081         0.102           9         0.081         0.102           9         0.081         0.102           9         0.081         0.102           9         0.081         0.102           9         0.081         0.102           9         0.081         0.102           9         0.081         0.102           8         0.072         0.094           8         0.072         0.094           8         0.072         0.094           9         0.032         0.101           8         0.072         0.034           9         0.033         0.0102           8         0.033         0.034
A3 A3 A3 A3 A3 A3 A3 A3 A3 A3 A3 A3 A3 A
Total Relations matrix         IV         IV<
tactors <th< th=""> <th< th="">          &lt;</th<></th<>

www

	А	В	С	D	E	525
A B C D E	0.186 0.230 0.237 0.237 0.232	0.228 0.176 0.222 0.219 0.202	0.256 0.260 0.204 0.234 0.247	0.165 0.167 0.168 0.136 0.183	0.165 0.167 0.168 0.173 0.136	<b>Table AIII.</b> Normalized Td Matrix for the dimension



MEQ 31,3         5226           5         11         <
<b>5226 5226 526 527 526 527 526 527</b>
2526         11         11         18         18         14         12         13         1
2526         M         B         B         C         C         C         C         C         C         C         C         C         C         C         C         C         C         D <thd< th=""> <thd< th=""> <thd< th=""> <thd< th=""></thd<></thd<></thd<></thd<>
M         B1         B2         B3         M         C1         C3         C3         C4         D1         D2         D3           255         0.258         0.288         0.288         0.142         0.143         0.113         0.114         0.284         0.294         <
M         B1         R2         B3         B4         C1         C2         C3         C4         C5         D1         D2           255         0.258         0.288         0.289         0.147         0.140         0.125         0.135         0.236
M         B1         R2         B3         B4         C1         C3         C4         C3         C4         C3           255         0.256         0.256         0.256         0.256         0.266
Mi         Bi         Bc         Bi         C         C         C         C         C           255         0256         0258         0258         0147         0140         0152         0136         0136           255         0256         0258         0258         0145         0142         0142         0142         0142         0142         0143         0146
AI         BI         B2         B3         B4         C1         C2         C3         C4         C5         C6           255         0258         0258         0258         0145         0142         0142         0143         0143         0143         0143         0143         0146         0132         0156         013
AI         B1         B2         B3         B4         C1         C2         C3         C4         C5           2361         0.258         0.258         0.258         0.258         0.147         0.140         0.152         0.138           257         0.266         0.258         0.258         0.166         0.176         0.133         0.136           257         0.266         0.258         0.258         0.144         0.142         0.133         0.133         0.133           256         0.256         0.258         0.258         0.144         0.143         0.146         0.133         0.133         0.133         0.133         0.133         0.133         0.134         0.134         0.134         0.134         0.134         0.134         0.134         0.134         0.134         0.134         0.134         0.134         0.134         0.134         0.134         0.134         0.134         0.134         0.134         0.133         0.133         0.133         0.133         0.133         0.134         0.134         0.134         0.134         0.134         0.134         0.134         0.134         0.134         0.134         0.134         0.134         0.134         0.134<
AI         BI         R2         B3         B4         CI         C2         C3         C4           3301         0.258         0.258         0.258         0.258         0.256         0.126         0.135
A4         B1         B2         B3         B4         C1         C2         C3           330         0.258         0.258         0.258         0.147         0.147         0.146         0.147         0.146           2287         0.258         0.258         0.258         0.256         0.2147         0.146         0.147         0.146         0.146         0.147         0.146
A4         B1         B2         B3         B4         C1         C2           2301         0.258         0.258         0.258         0.256         0.145         0.145           286         0.258         0.258         0.256         0.144         0.145         0.145           287         0.256         0.256         0.266         0.144         0.146         0.146           286         0.255         0.266         0.231         0.166         0.144         0.146           287         0.256         0.266         0.213         0.143         0.146         0.146           286         0.256         0.266         0.213         0.144         0.175         0.144           287         0.266         0.236         0.144         0.143         0.146         0.144           286         0.331         0.166         0.144         0.143         0.144         0.145           286         0.336         0.114         0.175         0.144         0.145         0.144           286         0.3317         0.266         0.144         0.144         0.145         0.144           286         0.3317         0.266         0.144
A4     B1     B2     B3     B4     C1       3301     0.258     0.258     0.258     0.256     0.147       2287     0.286     0.286     0.286     0.186     0.147       2386     0.258     0.286     0.286     0.266     0.147       247     0.266     0.266     0.266     0.146     0.147       257     0.275     0.266     0.213     0.166     0.147       256     0.256     0.266     0.213     0.166     0.147       256     0.256     0.266     0.236     0.1146     0.161       257     0.256     0.266     0.236     0.1146     0.161       256     0.256     0.266     0.236     0.1146     0.161       256     0.256     0.236     0.1179     0.166     0.145       257     0.256     0.236     0.1179     0.176     0.166       256     0.294     0.217     0.179     0.166     0.161       256     0.294     0.217     0.266     0.216     0.161       256     0.294     0.216     0.226     0.114     0.126       256     0.294     0.217     0.266     0.266     0.114       256 </th
A4         B1         B2         B3         B4           2301         0.258         0.258         0.299         0.166           2285         0.283         0.266         0.216         0.236           2301         0.258         0.256         0.266         0.116           2285         0.283         0.266         0.216         0.236           2305         0.256         0.256         0.236         0.116           2305         0.235         0.296         0.116         0.235           2305         0.235         0.236         0.236         0.136           2305         0.317         0.286         0.316         0.216           2306         0.317         0.286         0.136         0.216           2317         0.286         0.316         0.216         0.236           2317         0.286         0.316         0.316         0.236           2317         0.286         0.316         0.316         0.177           2317         0.286         0.286         0.114         0.178           2317         0.286         0.286         0.128         0.178           2316         0.286         <
A4         B1         B2         B3           .301         0.258         0.268         0.299           .287         0.268         0.268         0.266           .287         0.268         0.268         0.266           .287         0.266         0.266         0.266           .288         0.266         0.266         0.266           .288         0.266         0.266         0.266           .289         0.266         0.266         0.266           .289         0.266         0.266         0.266           .289         0.313         0.166         0.316           .273         0.266         0.266         0.266           .286         0.3316         0.266         0.266           .287         0.266         0.266         0.266           .286         0.266         0.276         0.266           .286         0.266         0.276         0.266           .286         0.286         0.266         0.266           .286         0.286         0.266         0.266           .286         0.286         0.286         0.266      /td>         .286         0.286         0.26
A4         B1         B2           .301         0.258         0.268           .287         0.268         0.268           .286         0.268         0.268           .287         0.266         0.256           .206         0.255         0.266           .273         0.266         0.255           .2897         0.266         0.256           .2896         0.255         0.266           .273         0.296         0.255           .2897         0.256         0.256           .2806         0.3313         0.2166           .2773         0.294         0.273           .2806         0.3317         0.266           .2771         0.296         0.276           .2773         0.294         0.271           .286         0.3317         0.266           .2773         0.294         0.271           .286         0.327         0.266           .2773         0.294         0.327           .278         0.296         0.226           .279         0.296         0.227           .286         0.237         0.237           .2728
A4 B1 3301 0.258 0.283 2.287 0.266 2.297 0.266 0.313 2.266 0.313 2.266 0.313 2.266 0.313 2.266 0.333 2.266 0.335 2.266 0.333 2.266 0.333 2.260 0.326 2.260 0.333 2.260 0.326 2.260 0.333 2.260 0.326 2.260 0.326 2.260 0.326 2.260 0.326 2.260 0.326 2.260 0.326 2.260 0.326 2.260 0.326 2.260 0.360 0
$A_4$ $A_4$ $A_4$ $A_5$ $B_5$
000000000000000000000000000000000000000
$\begin{array}{c} {\rm A3} \\ {\rm A3} \\ {\rm A3} \\ {\rm 0.252} \\ {\rm 0.217} \\ {\rm 0.2112} \\ {\rm 0.221} \\ {\rm 0.22$
$\begin{array}{c} \mathrm{A2} \\ \mathrm{A3} \\$
Table AIV.         Image: Wight and Wight an
for the dimension         국왕영본교路路초고강양강양양도금음음침교요요

8	0.300 0.329 0.109 0.262 0.267 0.267 0.267 0.267 0.164 0.113 0.135 0.135 0.135 0.135 0.135 0.135 0.137 0.137 0.137 0.137 0.137 0.137 0.137 0.137 0.137 0.137 0.137 0.137 0.137 0.137 0.137 0.137 0.136 0.137 0.136 0.136 0.137 0.137 0.136 0.1377 0.13777 0.13777 0.137777 0.13777777777777777777777777777777777777
E3	$\begin{array}{c} 0.301\\ 0.272\\ 0.137\\ 0.296\\ 0.239\\ 0.239\\ 0.239\\ 0.239\\ 0.239\\ 0.239\\ 0.261\\ 0.110\\ 0.165\\ 0.115\\ 0.116\\ 0.116\\ 0.116\\ 0.116\\ 0.171\\ 0.241\\ 0.211\\ 0.211\\ 0.211\\ 0.211\\ 0.211\\ 0.211\\ 0.211\\ 0.212\\ 0.211\\ 0.211\\ 0.212\\ 0.222\\ 0.212\\ 0.222\\ 0.$
E	$\begin{array}{c} 0.240\\ 0.240\\ 0.0357\\ 0.0357\\ 0.0204\\ 0.221\\ 0.221\\ 0.128\\$
2	0.248 0.318 0.248 0.248 0.185 0.257 0.257 0.108 0.116 0.117 0.115 0.115 0.115 0.115 0.115 0.115 0.116 0.116 0.116 0.252 0.262 0.262 0.262 0.262 0.262 0.262 0.262 0.262 0.262 0.262 0.264 0.264 0.264 0.276 0.276 0.276 0.276 0.276 0.276 0.276 0.276 0.277 0.277 0.277 0.277 0.277 0.277 0.277 0.277 0.277 0.277 0.277 0.277 0.277 0.276 0.276 0.277 0.277 0.277 0.277 0.277 0.277 0.277 0.277 0.277 0.277 0.277 0.277 0.277 0.277 0.277 0.277 0.277 0.277 0.276 0.276 0.276 0.276 0.276 0.276 0.276 0.276 0.276 0.276 0.276 0.276 0.276 0.276 0.276 0.276 0.276 0.276 0.277 0.277 0.277 0.277 0.277 0.277 0.277 0.277 0.277 0.2766 0.276 0.276 0.276 0.276 0.276 0.276 0.2766 0.276 0.276 0.276 0.27
E E	0.218 0.228 0.228 0.278 0.278 0.278 0.139 0.139 0.139 0.139 0.125 0.1570
23	$\begin{array}{c} 0.235\\ 0.2281\\ 0.2281\\ 0.2295\\ 0.2295\\ 0.2295\\ 0.2295\\ 0.2295\\ 0.2295\\ 0.2342\\ 0.145\\ 0.146\\ 0.146\\ 0.146\\ 0.146\\ 0.146\\ 0.107\\ 0.146\\ 0.134\\ 0.107\\ 0.247\\ 0.2364\\ 0.2$
E	$\begin{array}{c} 0.225\\ 0.228\\ 0.227\\ 0.271\\ 0.277\\ 0.229\\ 0.178\\ 0.114\\ 0.144\\ 0.144\\ 0.144\\ 0.132\\ 0.132\\ 0.132\\ 0.132\\ 0.161\\ 0.$
5	$\begin{array}{c} 0.249\\ 0.229\\ 0.236\\ 0.236\\ 0.253\\ 0.253\\ 0.253\\ 0.253\\ 0.253\\ 0.253\\ 0.253\\ 0.253\\ 0.230\\ 0.151\\ 0.1101\\ 0.1101\\ 0.1130\\ 0.1130\\ 0.1130\\ 0.1130\\ 0.1141\\ 0.1130\\ 0.1130\\ 0.1141\\ 0.1141\\ 0.1141\\ 0.1141\\ 0.116$
C	$\begin{array}{c} 0.278\\ 0.269\\ 0.301\\ 0.152\\ 0.320\\ 0.220\\ 0.254\\ 0.159\\ 0.159\\ 0.1166\\ 0.165\\ 0.165\\ 0.165\\ 0.165\\ 0.166\\ 0.165\\ 0.166\\ 0.166\\ 0.166\\ 0.163\\ 0.163\\ 0.0374\\ 0.245\\ 0.0374\\ 0.245\\ 0.0374\\ 0.245\\ 0.0378\\ 0.036\\ 0.0374\\ 0.0378\\ 0.036\\ 0.036\\ 0.0378\\ 0.036\\ 0$
පි	$\begin{array}{c} 0.233\\ 0.268\\ 0.244\\ 0.256\\ 0.294\\ 0.294\\ 0.298\\ 0.301\\ 0.116\\ 0.116\\ 0.116\\ 0.116\\ 0.118\\ 0.116\\ 0.118\\ 0.118\\ 0.118\\ 0.118\\ 0.118\\ 0.118\\ 0.118\\ 0.118\\ 0.118\\ 0.118\\ 0.206\\ 0.270\\ 0.226\\ 0.270\\ 0.266\\ 0.270\\ 0.266\\ 0.270\\ 0.266\\ 0.$
2	$\begin{array}{c} 0.282\\ 0.268\\ 0.312\\ 0.137\\ 0.137\\ 0.137\\ 0.249\\ 0.249\\ 0.029\\ 0.079\\ 0.157\\ 0.0156\\ 0.157\\ 0.0156\\ 0.0156\\ 0.0157\\ 0.005\\ 0.00$
3	$\begin{array}{c} 0.259\\ 0.295\\ 0.295\\ 0.295\\ 0.295\\ 0.295\\ 0.295\\ 0.295\\ 0.295\\ 0.295\\ 0.268\\ 0.1162\\ 0.1164\\ 0.1164\\ 0.1172\\ 0.1188\\ 0.1172\\ 0.1188\\ 0.1172\\ 0.1188\\ 0.1172\\ 0.1188\\ 0.1172\\ 0.1188\\ 0.1172\\ 0.1188\\ 0.1172\\ 0.1188\\ 0.1172\\ 0.1188\\ 0.1172\\ 0.1268\\ 0.$
8	0.233 0.277 0.277 0.277 0.277 0.275 0.265 0.256 0.113 0.113 0.1142 0.1142 0.1142 0.1142 0.1142 0.1142 0.1142 0.1142 0.1142 0.1142 0.1160 0.1160 0.1160 0.1160 0.1160 0.1160 0.1160 0.1160 0.1160 0.1160 0.1160 0.1160 0.1170 0.1160 0.11700 0.11700 0.11700 0.11700 0.11700 0.11700 0.1170000000000
5	$\begin{array}{c} 0.247\\ 0.257\\ 0.199\\ 0.297\\ 0.296\\ 0.226\\ 0.226\\ 0.176\\ 0.147\\ 0.147\\ 0.147\\ 0.147\\ 0.147\\ 0.147\\ 0.147\\ 0.133\\ 0.133\\ 0.133\\ 0.133\\ 0.133\\ 0.133\\ 0.133\\ 0.147\\ 0.147\\ 0.147\\ 0.147\\ 0.147\\ 0.147\\ 0.147\\ 0.126\\ 0.026\\ 0.$
X I	$\begin{array}{c} 0.201 \\ 0.288 \\ 0.288 \\ 0.284 \\ 0.282 \\ 0.282 \\ 0.282 \\ 0.282 \\ 0.282 \\ 0.0281 \\ 0.0114 \\ 0.0114 \\ 0.0113 \\ 0.0114 \\ 0.0113 \\ 0.0113 \\ 0.0113 \\ 0.0117 \\ 0.0113 \\ 0.0117 \\ 0.0117 \\ 0.0117 \\ 0.0117 \\ 0.0262 \\ 0.0326 \\ 0.032$
8	0.227 0.231 0.237 0.235 0.2355 0.2355 0.2355 0.2355 0.177 0.02555 0.0255 0.02550 0.02550 0.02550 0.02550 0.02550 0.02550 0.02550 0.02550 0.02550000000000
8	$\begin{array}{c} 0.236\\ 0.277\\ 0.269\\ 0.269\\ 0.313\\ 0.0313\\ 0.0313\\ 0.0316\\ 0.316\\ 0.316\\ 0.316\\ 0.316\\ 0.316\\ 0.316\\ 0.316\\ 0.174\\ 0.074\\ 0.077\\ 0.290\\ 0.172\\ 0.226\\$
E E	0.236 0.276 0.277 0.267 0.267 0.177 0.177 0.145 0.145 0.148 0.148 0.148 0.148 0.148 0.148 0.148 0.148 0.142 0.133 0.142 0.142 0.133 0.142 0.142 0.142 0.142 0.142 0.142 0.142 0.142 0.142 0.142 0.142 0.142 0.142 0.142 0.142 0.142 0.142 0.142 0.1440000000000
A4	0.260 0.321 0.218 0.202 0.254 0.254 0.254 0.143 0.143 0.143 0.146 0.143 0.146 0.143 0.146 0.143 0.146 0.143 0.146 0.1660
A3	0.259 0.303 0.152 0.152 0.287 0.286 0.266 0.168 0.168 0.168 0.168 0.168 0.153 0.153 0.153 0.153 0.153 0.153 0.153 0.154 0.131 0.335 0.335 0.335 0.335 0.326 0.226 0.2270 0.2270 0.226 0.2270 0.2270 0.226 0.2270 0.226 0.2270 0.2270 0.2270 0.2270 0.2270 0.2270 0.2270 0.2270 0.2270 0.2270 0.2270 0.2270 0.2270 0.2270 0.2270 0.2270 0.22700 0.22700 0.22700 0.22700 0.22700 0.2270000000000
A2	0.257 0.193 0.255 0.255 0.255 0.255 0.255 0.1142 0.116 0.1142 0.116 0.1156 0.116 0.1156 0.1156 0.1156 0.1156 0.1156 0.1156 0.1156 0.1156 0.1156 0.1156 0.1156 0.1156 0.1157 0.1259 0.1259 0.1259 0.1259 0.1259 0.1259 0.1259 0.1259 0.1259 0.1259 0.1259 0.1259 0.1256 0.1255 0.1256 0.1255 0.1255 0.1255 0.1255 0.1255 0.1255 0.1255 0.1255 0.12566 0.1256 0.1256 0.1256 0.1256 0.12566 0.12566 0.12566 0
AI	0.165 0.228 0.228 0.258 0.258 0.258 0.258 0.258 0.258 0.147 0.147 0.147 0.147 0.1280 0.128
	BEED500000000000000000000000000000000000
للاستشارات	ill

Transition to circular economy via DANP

Table AV.Unweighted matrix(W)

MEQ	E3	$\begin{array}{c} 0.070\\ 0.076\\ 0.076\\ 0.071\\ 0.071\\ 0.071\\ 0.073\\ 0.033\\ 0.$
01,0	E2	$\begin{array}{c} 0.070\\ 0.063\\ 0.063\\ 0.069\\ 0.069\\ 0.048\\ 0.041\\ 0.024\\ 0.041\\ 0.025\\ 0.036\\ 0.036\\ 0.036\\ 0.036\\ 0.036\\ 0.056\\ 0.$
	El	0.056 0.071 0.071 0.083 0.043 0.043 0.043 0.032 0.0330 0.033 0.0330 0.0330 0.0330 0.0330 0.0330 0.0330 0.0330 0.03300000000
528	D4	0.059 0.075 0.071 0.071 0.071 0.072 0.035 0.025 0.025 0.025 0.026 0.026 0.027 0.027 0.028 0.0028 0.0008 0.0008 0.0008 0.0008 0.0008 0.0008 0.000800000000
	D3	0.052 0.065 0.065 0.066 0.066 0.064 0.061 0.063 0.032 0.032 0.033 0.037 0.032 0.037 0.037 0.037 0.037 0.037 0.037 0.037 0.037 0.037 0.037 0.037 0.037 0.037 0.037 0.037 0.032 0.037 0.032 0.037 0.0320 0.0320 0.0320 0.0320000000000
	D2	$\begin{array}{c} 0.056\\ 0.057\\ 0.067\\ 0.063\\ 0.053\\ 0.056\\ 0.066\\ 0.034\\ 0.024\\ 0.047\\ 0.042\\ 0.042\\ 0.042\\ 0.042\\ 0.042\\ 0.038\\ 0.038\\ 0.038\\ 0.038\\ 0.038\\ 0.038\\ 0.038\\ 0.038\\ 0.038\\ 0.038\\ 0.050\\ 0.$
	DI	0.053 0.067 0.067 0.058 0.058 0.0330 0.0330 0.0330 0.0330 0.0330 0.0330 0.0330 0.0330 0.0330 0.03300000000
	C7	0.059 0.056 0.056 0.056 0.056 0.068 0.068 0.068 0.068 0.025 0.0205 0.025 0.0205 0.0205 0.0205 0.0205 0.0205 0.0205 0.0200 0.0200 0.0200 0.0200 0.021 0.021 0.025 0.0200 0.021 0.021 0.025 0.0200 0.0210 0.0210 0.0210 0.0210 0.0210 0.0210 0.0210 0.0210 0.0210 0.02100 0.02100 0.0210000000000
	C6	0.066 0.071 0.071 0.071 0.073 0.073 0.073 0.073 0.073 0.073 0.073 0.073 0.073 0.073 0.073 0.073 0.073 0.073 0.073 0.073 0.073 0.071 0.073 0.071 0.0736 0.071 0.0736 0.071 0.0736 0.071 0.0736 0.0771 0.0772 0.0772 0.0771 0.07720 0.07720 0.07720000000000
	C5	0.055 0.064 0.068 0.068 0.061 0.065 0.067 0.067 0.023 0.023 0.023 0.023 0.023 0.023 0.023 0.023 0.023 0.023 0.023 0.023 0.040 0.055 0.040 0.055 0.040 0.055 0.055 0.055 0.055 0.055 0.065 0.005 0.065 0.0050 0.00500000000
	C4	$\begin{array}{c} 0.067\\ 0.064\\ 0.074\\ 0.073\\ 0.055\\ 0.055\\ 0.055\\ 0.055\\ 0.032\\ 0.032\\ 0.032\\ 0.032\\ 0.032\\ 0.032\\ 0.032\\ 0.032\\ 0.032\\ 0.032\\ 0.032\\ 0.032\\ 0.032\\ 0.032\\ 0.032\\ 0.032\\ 0.032\\ 0.032\\ 0.032\\ 0.040\\ 0.040\\ 0.040\\ 0.040\\ 0.040\\ 0.064\\ 0.066\\ 0.040\\ 0.064\\ 0.066\\ 0.064\\ 0.066\\ 0.$
	ខ	0.061 0.0360 0.0360 0.0770 0.0721 0.0725 0.07550 0.07550 0.07550 0.07550 0.07550 0.07550000000000
	C2	0.055 0.065 0.065 0.065 0.065 0.065 0.071 0.071 0.071 0.072 0.023 0.025 0.027 0.025 0.027 0.027 0.025 0.027 0.025 0.027 0.025 0.027 0.025 0.027 0.025 0.027 0.007 0.0270 0.0270 0000000000
	CI	0.059 0.071 0.071 0.071 0.072 0.073 0.039 0.039 0.033 0.033 0.033 0.033 0.033 0.033 0.033 0.033 0.033 0.033 0.033 0.033 0.033 0.035 0.035 0.035 0.035 0.035 0.037 0.007 0.0370000000000
	B4	0.046 0.065 0.065 0.065 0.065 0.065 0.047 0.047 0.021 0.021 0.023 0.047 0.055 0.065 0.065 0.065 0.065 0.065 0.065 0.065 0.065 0.005 0.065 0.005 00000000
	B3	0.052 0.055 0.0055 0.0055 0.0050 0.060 0.046 0.046 0.046 0.052 0.052 0.052 0.053 0.052 0.053 0.052 0.053 0.052 0.052 0.053 0.052 0.052 0.052 0.052 0.052 0.055 0.052 0.0550 0.0550 0.0550 0.0550 0.0550 0.0550 0.0550 0.0550 0.0550 0.05500000000
	B2	$\begin{array}{c} 0.054\\ 0.064\\ 0.062\\ 0.050\\ 0.052\\ 0.055\\ 0.036\\ 0.037\\ 0.048\\ 0.048\\ 0.048\\ 0.048\\ 0.048\\ 0.048\\ 0.052\\ 0.052\\ 0.053\\ 0.048\\ 0.048\\ 0.048\\ 0.048\\ 0.069\\ 0.052\\ 0.069\\ 0.068\\ 0.$
	B1	0.054 0.063 0.061 0.061 0.061 0.033 0.033 0.034 0.0380 0.03800 0.03800 0.0380000000000
	A4	0.048 0.0560 0.0040 0.0037 0.058 0.0058 0.0041 0.037 0.0364 0.041 0.036 0.041 0.036 0.041 0.041 0.041 0.041 0.041 0.053 0.041 0.053 0.053 0.054 0.053 0.054 0.0550 0.0550 0.0550 0.0550 0.05500000000
	A3	0.048 0.058 0.053 0.053 0.061 0.061 0.049 0.043 0.044 0.033 0.044 0.033 0.044 0.033 0.044 0.033 0.044 0.033 0.044 0.033 0.044 0.057 0.057 0.057
	A2	0.050 0.036 0.047 0.065 0.065 0.065 0.038 0.041 0.041 0.043 0.043 0.043 0.043 0.043 0.043 0.043 0.043 0.043 0.043 0.043 0.041 0.043 0.054 0.071 0.071 0.071 0.072
Table AVI.		0.031 0.057 0.042 0.042 0.056 0.068 0.068 0.038 0.038 0.033 0.033 0.033 0.033 0.033 0.033 0.033 0.033 0.033 0.033 0.033 0.033 0.033 0.055 0.033 0.055 0.033 0.055 0.033 0.055 0.035 0.035 0.035 0.035 0.0550 0.0550 0.0550 0.0550 0.05500000000
Weighted matrix ( <i>W</i> *)		825557555555555555555555555555555555555
فسارة للاستشارات		
Table AVI. Weighted matrix (W*)	41	Al 0.031 A2 0.055 A3 0.042 A3 0.042 A3 0.042 A3 0.042 C 0.056 C 1 0.038 C 2 0.056 C 2 0.038 C 2

<u>蹈</u>	0.066 0.074 0.071 0.072 0.065 0.065 0.046 0.046 0.046 0.043 0.041 0.041	0.035 0.038 0.057 0.057 0.049 0.041 0.041 0.068 0.068 0.068 0.067	Transition
원	$\begin{array}{c} 0.065\\ 0.074\\ 0.076\\ 0.071\\ 0.071\\ 0.071\\ 0.065\\ 0.065\\ 0.066\\ 0.046\\ 0.043\\ 0.040\\ 0.041\\ 0.041\end{array}$	$\begin{array}{c} 0.035\\ 0.038\\ 0.057\\ 0.046\\ 0.049\\ 0.041\\ 0.059\\ 0.066\\ 0.066\\ 0.066\end{array}$	to circular economy
EI	0.066 0.074 0.075 0.071 0.072 0.072 0.065 0.065 0.065 0.046 0.043 0.041 0.041	0.035 0.038 0.057 0.046 0.049 0.041 0.060 0.067 0.067	V1a DANP
ž	$\begin{array}{c} 0.065\\ 0.074\\ 0.076\\ 0.071\\ 0.071\\ 0.071\\ 0.065\\ 0.066\\ 0.046\\ 0.046\\ 0.043\\ 0.041\\ 0.041\end{array}$	$\begin{array}{c} 0.035\\ 0.038\\ 0.057\\ 0.046\\ 0.049\\ 0.041\\ 0.059\\ 0.066\\ 0.066\\ 0.066\end{array}$	529
D3	$\begin{array}{c} 0.066\\ 0.074\\ 0.076\\ 0.071\\ 0.072\\ 0.072\\ 0.065\\ 0.066\\ 0.046\\ 0.046\\ 0.043\\ 0.041\\ 0.041\end{array}$	0.035 0.038 0.057 0.046 0.041 0.060 0.068 0.068 0.068	
D2	$\begin{array}{c} 0.066\\ 0.074\\ 0.076\\ 0.071\\ 0.072\\ 0.065\\ 0.066\\ 0.046\\ 0.041\\ 0.041\\ 0.041\end{array}$	$\begin{array}{c} 0.035\\ 0.036\\ 0.038\\ 0.057\\ 0.046\\ 0.041\\ 0.047\\ 0.060\\ 0.047\\ 0.068\\ 0.067\\ 0.068\\ 0.067\end{array}$	
ā	$\begin{array}{c} 0.066\\ 0.074\\ 0.071\\ 0.071\\ 0.072\\ 0.065\\ 0.046\\ 0.046\\ 0.043\\ 0.041\\ 0.041\end{array}$	$\begin{array}{c} 0.035\\ 0.038\\ 0.038\\ 0.046\\ 0.041\\ 0.041\\ 0.047\\ 0.068\\ 0.067\\ 0.067\\ 0.067\end{array}$	
5	$\begin{array}{c} 0.065\\ 0.074\\ 0.071\\ 0.071\\ 0.071\\ 0.065\\ 0.046\\ 0.046\\ 0.042\\ 0.041\end{array}$	$\begin{array}{c} 0.035\\ 0.038\\ 0.038\\ 0.046\\ 0.041\\ 0.041\\ 0.059\\ 0.047\\ 0.066\\ 0.067\\ 0.066\end{array}$	
ප	$\begin{array}{c} 0.066\\ 0.074\\ 0.076\\ 0.071\\ 0.071\\ 0.065\\ 0.066\\ 0.046\\ 0.043\\ 0.041\\ 0.041\end{array}$	$\begin{array}{c} 0.035\\ 0.036\\ 0.046\\ 0.041\\ 0.041\\ 0.047\\ 0.068\\ 0.047\\ 0.068\\ 0.066\\ 0.068\\ 0.068\end{array}$	
8	$\begin{array}{c} 0.066\\ 0.074\\ 0.076\\ 0.071\\ 0.072\\ 0.065\\ 0.066\\ 0.046\\ 0.041\\ 0.041\\ 0.041\end{array}$	$\begin{array}{c} 0.035\\ 0.036\\ 0.046\\ 0.041\\ 0.041\\ 0.047\\ 0.068\\ 0.047\\ 0.068\\ 0.067\\ 0.067\\ 0.061\end{array}$	
5	$\begin{array}{c} 0.066\\ 0.074\\ 0.071\\ 0.071\\ 0.071\\ 0.065\\ 0.066\\ 0.046\\ 0.040\\ 0.041\\ 0.041\\ 0.041\end{array}$	$\begin{array}{c} 0.035\\ 0.038\\ 0.057\\ 0.046\\ 0.041\\ 0.041\\ 0.060\\ 0.047\\ 0.068\\ 0.067\\ 0.067\\ 0.067\\ 0.067\\ 0.068\end{array}$	
ខ	$\begin{array}{c} 0.066\\ 0.074\\ 0.056\\ 0.071\\ 0.071\\ 0.071\\ 0.065\\ 0.068\\ 0.046\\ 0.040\\ 0.041\\ 0.041\end{array}$	$\begin{array}{c} 0.035\\ 0.038\\ 0.057\\ 0.046\\ 0.041\\ 0.060\\ 0.047\\ 0.068\\ 0.067\\ 0.067\\ 0.067\\ 0.067\end{array}$	
ප	$\begin{array}{c} 0.066\\ 0.074\\ 0.056\\ 0.071\\ 0.072\\ 0.065\\ 0.068\\ 0.046\\ 0.041\\ 0.041\\ 0.041\end{array}$	$\begin{array}{c} 0.035\\ 0.038\\ 0.057\\ 0.046\\ 0.041\\ 0.060\\ 0.047\\ 0.068\\ 0.067\\ 0.067\\ 0.061\end{array}$	
5	$\begin{array}{c} 0.066\\ 0.074\\ 0.071\\ 0.071\\ 0.072\\ 0.072\\ 0.065\\ 0.065\\ 0.046\\ 0.040\\ 0.041\\ 0.041\end{array}$	$\begin{array}{c} 0.035\\ 0.038\\ 0.057\\ 0.046\\ 0.041\\ 0.060\\ 0.047\\ 0.068\\ 0.067\\ 0.068\\ 0.061\\ 0.067\\ 0.061\\ 0.$	
5 2 2	$\begin{array}{c} 0.066\\ 0.074\\ 0.071\\ 0.072\\ 0.065\\ 0.065\\ 0.068\\ 0.046\\ 0.040\\ 0.041\\ 0.041\end{array}$	$\begin{array}{c} 0.035\\ 0.038\\ 0.057\\ 0.046\\ 0.041\\ 0.041\\ 0.060\\ 0.047\\ 0.068\\ 0.067\\ 0.067\\ 0.067\\ 0.061\\ 0.061\end{array}$	
B	$\begin{array}{c} 0.066\\ 0.074\\ 0.057\\ 0.057\\ 0.072\\ 0.072\\ 0.065\\ 0.065\\ 0.069\\ 0.047\\ 0.041\\ 0.041\\ 0.041\end{array}$	$\begin{array}{c} 0.035\\ 0.038\\ 0.058\\ 0.046\\ 0.041\\ 0.041\\ 0.060\\ 0.047\\ 0.068\\ 0.067\\ 0.067\\ 0.067\\ 0.061\\ 0.061\end{array}$	
B3	$\begin{array}{c} 0.066\\ 0.074\\ 0.057\\ 0.057\\ 0.072\\ 0.072\\ 0.066\\ 0.066\\ 0.067\\ 0.047\\ 0.041\\ 0.041\\ 0.041\end{array}$	$\begin{array}{c} 0.035\\ 0.038\\ 0.058\\ 0.049\\ 0.041\\ 0.041\\ 0.061\\ 0.067\\ 0.067\\ 0.067\\ 0.067\\ 0.067\\ 0.061\end{array}$	
E	0.066 0.074 0.075 0.071 0.072 0.065 0.065 0.068 0.046 0.043 0.041 0.041	$\begin{array}{c} 0.035\\ 0.038\\ 0.057\\ 0.049\\ 0.049\\ 0.041\\ 0.061\\ 0.067\\ 0.067\\ 0.067\\ 0.067\\ 0.061\\ 0.061\end{array}$	
A4	0.066 0.074 0.071 0.071 0.072 0.065 0.065 0.068 0.068 0.046 0.040 0.041 0.041	$\begin{array}{c} 0.035\\ 0.038\\ 0.057\\ 0.049\\ 0.049\\ 0.041\\ 0.061\\ 0.067\\ 0.067\\ 0.067\\ 0.067\\ 0.061\\ 0.061\end{array}$	ower 15
A3	0.066 0.074 0.071 0.071 0.071 0.071 0.065 0.065 0.065 0.065 0.046 0.040 0.041 0.041	$\begin{array}{c} 0.035\\ 0.038\\ 0.057\\ 0.046\\ 0.049\\ 0.041\\ 0.057\\ 0.047\\ 0.066\\ 0.066\\ 0.066\end{array}$	ed to po
A2	$\begin{array}{c} 0.066\\ 0.074\\ 0.076\\ 0.071\\ 0.072\\ 0.065\\ 0.065\\ 0.066\\ 0.046\\ 0.040\\ 0.041\\ 0.041\end{array}$	$\begin{array}{c} 0.035\\ 0.038\\ 0.057\\ 0.046\\ 0.041\\ 0.041\\ 0.067\\ 0.067\\ 0.067\\ 0.067\\ 0.067\\ 0.067\\ 0.061\\ 0.061\end{array}$	W* rais
Ŧ	$\begin{array}{c} 0.066\\ 0.074\\ 0.076\\ 0.072\\ 0.072\\ 0.065\\ 0.065\\ 0.066\\ 0.046\\ 0.040\\ 0.041\\ 0.041\\ 0.041\end{array}$	$\begin{array}{c} 0.035\\ 0.038\\ 0.057\\ 0.046\\ 0.049\\ 0.041\\ 0.061\\ 0.067\\ 0.067\\ 0.067\\ 0.067\\ 0.061\\ 0.061\end{array}$	Table AVII.
	3325 <u>8</u> 888 <u>8</u> 582	8855555555555 8655555555555555555555555	2
	i		
	P1		WV

Reproduced with permission of copyright owner. Further reproduction prohibited without permission.

